Author's response to reviews

Title: Prevalence of chronic fatigue syndrome in metropolitan, urban, and rural Georgia

Authors:

William C Reeves (wcr1@cdc.gov)
James F Jones (jai9@cdc.gov)
Elizabeth Maloney (evm3@cdc.gov)
Christine Heim (cmheim@emory.edu)
David C Hoaglin (dave_hoaglin@abtassoc.com)
Roumiana S Boneva (rrb5@cdc.gov)
Marjorie Morrissey (Marjorie_Morrissey@abtassoc.com)
Rebecca Devlin (Rebecca_Devlin@abtassoc.com)

Version: 5 Date: 29 May 2007

Author's response to reviews:

I have made the changes requested by the editorial team. However, I did not understand the last request "Please put the heading 'Tables' before the tables." I interpreted this literally and included the heading Tables on the line before Table 1.